Author's response to reviews

Title: Elevated central venous pressure is associated with impairment of microcirculatory blood flow in sepsis: a post hoc analysis

Authors:

Namkje N.A.R. Vellinga (namkjevellinga@hotmail.com)
Can C. Ince (c.ince@erasmusmc.nl)
Christiaan E.C. Boerma (e.boerma@chello.nl)

Version: 2 Date: 22 January 2013

Author's response to reviews:

Dear Sir, dear Madam,

Corrections regarding the position of the table/figure legend were made, as well as an adaptation to the text regarding the ethics committee/informed consent.

Sincerely,

Namkje Vellinga